UCSF UC San Francisco Previously Published Works

Title

Impact of extended-course oral nirmatrelvir/ritonavir in established Long COVID: a case series.

Permalink

https://escholarship.org/uc/item/8598s38f

Journal Communications Medicine, 4(1)

ISSN

2730-664X

Authors

Cohen, Alison Jaudon, Toni Schurman, Eric <u>et al.</u>

Publication Date

2025

DOI

10.1038/s43856-024-00668-8

Peer reviewed

6

https://doi.org/10.1038/s43856-024-00668-8

Impact of extended-course oral nirmatrelvir/ritonavir in established Long COVID: a case series

Check for updates

Alison K. Cohen **D**¹ \boxtimes , Toni Wall Jaudon², Eric M. Schurman **D**³, Lisa Kava⁴, Julia Moore Vogel **D**⁵, Julia Haas-Godsil⁶, Daniel Lewis⁷, Samantha Crausman⁸, Kate Leslie⁹, Siobhan Christine Bligh¹⁰, Gillian Lizars¹¹, JD Davids¹², Saniya Sran¹³, Michael Peluso¹⁴ & Lisa McCorkell **D**¹⁵

Abstract

Background Prior case series suggest that a 5-day course of oral Paxlovid (nirmatrelvir/ ritonavir) benefits some people with Long COVID, within and/or outside of the context of an acute reinfection. To the best of our knowledge, there have been no prior case series of people with Long COVID who have attempted longer courses of nirmatrelvir/ritonavir. **Methods** We documented a case series of 13 individuals with Long COVID who initiated extended courses (>5 days; range: 7.5–30 days) of oral nirmatrelvir/ritonavir outside (n = 11) of and within (n = 2) the context of an acute SARS-CoV-2 infection. Participants reported on symptoms and health experiences before, during, and after their use of nirmatrelvir/ritonavir. **Results** Among those who take an extended course of nirmatrelvir/ritonavir outside of the context of an acute infection, some experience a meaningful reduction in symptoms, although not all benefits persist. Others experience no effect on symptoms. One participant stopped early due to intense stomach pain. For the two participants who took an extended course of nirmatrelvir/ritonavir within the context of an acute reinfection, both report eventually returning to their pre-re-infection baseline.

Conclusions Extended courses of nirmatrelvir/ritonavir may have meaningful benefits for some people with Long COVID but not others. We encourage researchers to study how and why nirmatrelvir/ritonavir benefits some and what course length is most effective, with the goal of informing clinical recommendations for using nirmatrelvir/ritonavir and/or other antivirals as a potential treatment for Long COVID.

Long COVID is an often debilitating, multisystemic illness that occurs following a SARS-CoV-2 infection¹ that affects \geq 76 million people worldwide^{2–4}. It can last for months, years, or potentially a lifetime. Over 1% of all American adults currently have severe activity limitations due to Long COVID⁵. Long COVID symptoms can include brain fog, fatigue, postexertional malaise (PEM), autonomic dysfunction, headache, tinnitus,

Plain language summary

Long COVID is an infection-associated chronic condition in which symptoms persist more than 12 weeks after an acute COVID-19 infection. Prior reports suggest that a 5-day course of oral Paxlovid (nirmatrelvir/ritonavir) may help some people with Long COVID, but there have not yet been any reports of people with Long COVID trying extended courses of nirmatrelvir/ritonavir. Our patient-led study documents the experiences of 13 people with Long COVID who tried extended courses of Paxlovid; some were in the midst of a reinfection and others were not. Some participants reported meaningful benefits, although not all improvements lasted longterm. Others experienced no changes in symptoms. Our results suggest that extended courses of nirmatrelvir/ritonavir could be beneficial for some people with Long COVID.

blurred vision, loss or alteration of smell or taste, palpitations, gastrointestinal symptoms, changes in sexual desire or capacity, menstrual cycle impacts, chronic cough, and chest pain^{6,7}.

The pathophysiology underlying Long COVID is an area of major ongoing investigation. After acute COVID-19, some patients do not fully clear the SARS-CoV-2 virus, and viral reservoirs may persist over long

¹Department of Epidemiology & Biostatistics and Philip R. Lee Institute for Health Policy Studies, School of Medicine, University of California San Francisco and Patient-Led Research Collaborative, San Francisco, CA, USA. ²Patient-Led Research Collaborative & University of Arkansas for Medical Sciences, Little Rock, AR, USA. ³Independent scholar, Seattle, WA, USA. ⁴Patient-Led Research Collaborative, Brisbane, QLD, Australia. ⁵Scripps Research Translational Institute and Patient-Led Research Collaborative, La Jolla, CA, USA. ⁶Independent scholar, Stamford, CT, USA. ⁷Patient-Led Research Collaborative, Campbell, CA, USA. ⁸Independent scholar, Rehoboth, MA, USA. ⁹Independent scholar, Boulder, CO, USA. ¹⁰Independent scholar, London, UK. ¹¹Independent scholar, Oakland, CA, USA. ¹²Strategies for High Impact/Long COVID Justice, Patient-Led Research Collaborative, New York City, NY, USA. ¹³Independent scholar, Los Angeles, CA, USA. ¹⁴Division of HIV, Infectious Diseases, and Global Medicine, University of California San Francisco, San Francisco, CA, USA. ¹⁵Patient-Led Research Collaborative, Oakland, CA, USA. ^{Senand} and Senand CA, USA. ¹⁶Patient-Led Research periods of time⁸⁻¹². For example, SARS-CoV-2 viral RNA and protein can persist in blood and/or tissue beyond the acute phase of infection¹³.

The oral antiviral Paxlovid (nirmatrelvir/ritonavir), when administered for acute COVID-19, can reduce viral activity¹⁴, symptom intensity, fatalities¹⁵, and potentially also the likelihood of developing Long COVID (current evidence is mixed^{16,17}). Prior case series^{18,19} suggested symptomatic improvement in people with Long COVID who completed a 5-day course of nirmatrelvir/ritonavir for symptom management or in response to a subsequent COVID-19 infection. For some cases, symptoms recurred following completion of treatment, suggesting that longer courses may be needed.

Here, we report 13 cases of people with Long COVID who took >5-day courses of nirmatrelvir/ritonavir and who described their lived experience of the impact of this agent on their Long COVID illness trajectory. We find that extended courses of nirmatrelvir/ritonavir may have meaningful benefits for some people with Long COVID but not others. These cases emphasize the need to study longer courses of antiviral therapy for Long COVID and lend further support to the pursuit of clinical trials testing various durations of nirmatrelvir/ritonavir therapy in people experiencing Long COVID.

Methods

Case selection

We compiled cases identified using two different methods. First, all individuals who had previously shared aspects of their experience with nirmatrelvir/ritonavir in public forums led by and for people with Long COVID were invited to participate by one or more of the co-authors who were also patients with Long COVID and members of the forum(s); all but two of the individuals invited this way were eager to participate. Second, others heard about the project via word-of-mouth and contacted us to share their experience. Participants chose to either share pre-existing written text (e.g., text shared in public forums, notes they wrote to clinicians) or talk with one of the authors about their experience. As needed, participants answered follow-up questions in writing. All participants consented to the presentation of their cases in this manuscript, including the level of detail provided. As part of our data validation process, which also included review by an epidemiologist (Author 1) and a clinical researcher (Author 14), all participants had the opportunity to review and edit the written description of their experience.

Case eligibility

All individuals who reported a diagnosis of Long COVID and who reported taking >5 days of nirmatrelvir/ritonavir at some point following their diagnosis who made themselves known to the research team were included in this case series. COVID-19 tests were not always available, so, per the recommendations of others, we included people who had clinically confirmed cases of COVID-19 in addition to those who had positive test results. All patients had been evaluated by at least one health care provider at some point during their Long COVID disease experience, who was able to validate that their symptoms were likely due to Long COVID. Participants typically reported on their experiences within 1 month of completing their course of nirmatrelvir/ritonavir, drawing from contemporaneous notes and/or medical records documenting their experiences. Given the exploratory, retrospective nature of this study, we relied on open-ended text to understand any nuanced changes that may have occurred. Because participants were not asked to share any identifiable information, this study was selfcertified as exempt from University of California San Francisco Institutional Review Board review. To protect the anonymity of participants, data are not publicly available.

Inclusion and ethics

Inclusion and ethics statement: The research sought to be relevant to the affected community (patients with Long COVID) and has included patients with Long COVID throughout the research process. The roles and responsibilities were agreed upon amongst all collaborators, and capacity-

building was provided for co-authors who were patients with Long COVID who did not have prior research experience.

Reporting summary

Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

Results

We report on the results of 13 cases who had Long COVID at the time of taking nirmatrelvir/ritonavir. Cases 1–11 (Tables 1 and 2) took an extended course of nirmatrelvir/ritonavir while negative for SARS-CoV-2, outside of the context of an acute infection; Cases 12–13 (Table 3) did so while positive for SARS-CoV-2, in the context of an acute refinfection. We also report SARS-CoV-2 vaccination history; unless otherwise noted, vaccination did not affect patients' Long COVID symptoms and patients reported no more than temporary side effects.

Case #1

In March 2020, a 56-year-old man with no prior medical history developed symptoms consistent with mild COVID-19. Confirmatory testing was not available at the time. Initially, most symptoms resolved, except for ongoing diarrhea. In August 2020, he developed symptoms of Long COVID, including severe fatigue, hypersomnia, brain fog, exercise intolerance, PEM, loss of hearing, photosensitivity, headaches lasting up to a week at a time, symptoms of vocal cord dysfunction, insomnia, elevated heart rate when standing and upon minor exertion, balance disturbance, vestibular dysfunction, and joint pain. He completed the primary SARS-CoV-2 vaccine series in February/March 2021 (Spikevax) and booster doses in November 2021 (Spikevax), May 2022 (Comirnaty), and November 2022 (Comirnaty).

In September 2022, he completed a 5-day course of nirmatrelvir/ ritonavir outside of the context of an acute COVID-19 infection, with no change in symptoms. In December 2022, he completed a 15-day course of nirmatrelvir/ritonavir outside of the course of an acute COVID-19 infection. Concomitant medications included cetirizine, famotidine, nattokinase, serrapeptase, and vitamins B, C, and D3. He began experiencing some improvement in symptoms during days 5-10, with more substantial improvement after 10 days of treatment. Following the 15-day course, he reported feeling substantially more energetic and that his muscles no longer felt like "dead weights." He also reported improvement in his neurocognitive symptoms, including decreased photosensitivity, and improved clarity of thought, memory, and word-finding ability. One week after completing the course, he reported almost complete resolution of Long COVID symptoms. For example, he could walk continuously for 45 min and lightly jog 3-4 times for approximately 1 min during exercise, with no exacerbation of symptoms or PEM. He also noted that arthritis in his fingers resolved after this course. Although his symptoms recurred to a certain extent within four weeks of completing the nirmatrelvir/ritonavir course, he reported an improved baseline.

Case #2

In March 2022, a 45-year-old woman with no pertinent medical history developed mild COVID-19. The infection was confirmed by nucleic acid amplification testing. Prior to her infection, she was a physically active athlete and had received the primary two-shot series of COVID-19 vaccines (Spikevax) in January/February 2021, and a Spikevax booster in November 2021. About three weeks after her acute infection, she developed Long COVID symptoms, including orthostatic intolerance, tachycardia, short-term memory loss, word-finding difficulties, fatigue, PEM, and symptoms of mast cell activation syndrome (MCAS) including food intolerances, rashes, and headaches. She was diagnosed with MCAS and postural orthostatic tachycardia syndrome (POTS); she attained partial relief with beta blockers, although her blood pressure regularly fluctuated between 80–145 (systolic) and 60–80 (diastolic). When her blood pressure was low, she experienced weakness, shortness of breath, light-headedness, and episodes of fainting.

1 56 year man March 2020 original widtype Jul 2020 2021 (Bab, Mar, Nobi), March 2022 Jul 2020 2021 (Bab, Nobi), March 2022 Jul 2023	-	56 y.o. man	March 2020							of Paxlovid
2 45 vo. womany March 2022 omicron And 2022 control May 2022 2023 (Jan, Fab, Zo21) Fab 2023 (15 d) y y 3 25 vo. womany May 2022 omicron May 2022 222 (Jan, Fab, Xo1) Jan 2023 (10 d) y y y 4 F1 vo. many Mag 2022 omicron May 2022 222 (Jan, May, Nov) Mar 2023 (15 d) y y y 6 Undicioned March 2020 onigrand May 2020 Nar 2023 (15 d) y y y 7 30 vo. man May 2022 omicron Nar 2020 Nar 2023 (15 d) y y y 7 30 vo. man May 2022 omicron Nar 2022 (15 d) y y y y 7 30 vo. man May 2020 omicron Jul 2022 Sat 2023 (15 d) y y y y 7 30 vo. man May 2020 omicron Jul 2022 Sat 302 (15 d) y y y y				original wild type	Jul 2020	2021 (Feb, Mar, Nov 2022 (May, Nov)		х	×	
3 5 you woman May 2022 Omicron May 2022 Omicron May 2022 Omicron May 2022 Omicron May 2023 May	N	45 y.o. woman		omicron	Apr 2022	2021 (Jan, Feb, 202 2022 (Oct)		×	Х	Y
4 51 you man 4 wg 2022 0 mginal wg 2022 (kg mil) y wg 2022 (kg mil) y wg 2022 (kg mil) y wg 2023 (kg mil) <td>3</td> <td>25 y.o. woman</td> <td>May 2022</td> <td>omicron</td> <td>May 2022</td> <td>2021 (Apr, Apr, Nov)</td> <td></td> <td>У</td> <td></td> <td>У</td>	3	25 y.o. woman	May 2022	omicron	May 2022	2021 (Apr, Apr, Nov)		У		У
5 women, age undisciosed March 2020 original wind type Mudi type Mar 2020 5 vaccines beginning Jun 2023 (10 d) Y Y 7 30 you man March 2020 omicroni, Mid type Jun 2022 Sept 2022 (30 d) Y Y 7 30 you man March 2020 omicroni, Mid type Jul 2022 2021 (Mar, April, Nov) May 2023 (15 d) Y Y 7 30 you man May 2022, Mid type Jul 2022 2021 (Mar, April, Nov) May 2023 (15 d) Y Y 7 30 you man May 2023 (15 d) May 2023 (15 d) Y Y Y 7 30 you man May 2023 (15 d) May 2023 (15 d) Y Y Y 7 30 you man May 2023 (15 d) May 2023 (15 d) Y Y Y 7 May 2023 (15 d) Y Y Y Y Y 7 May 2023 (15 d) Y Y Y Y Y 7 May 2023 (15 d) Y Y Y	4	51 y.o. man	Aug 2022	omicron	Aug 2022	2021 (Apr, May, Nov 2022 (Aug)		×	~	
0 40 you March 2020 original wild type May 2020 Not specified Sept 2020 (30 d) Y Y 7 30 you May 2022 omlorony. July 2022 omlorony. July 2022 omlorony. July 2022 providuals indicated as hand specified inprovements. In the break are due to immarket/informatic valacyclorie, or both. July 2022 2021 (Mar. April, Nov) May 2023 (15 d) Y Y For individuals indicated as hand specified inprovements. In the break are due to immarket/informatic valacyclorie, or both. July 2022 (15 d) Y Y Initiation in the break are due to immarket/informatic valacyclorie, or both. S For individuals indicated as hand specified as hand specified in the break are due to immarket/informatic valacyclorie, or both. July 2022 July 2022 (15 d) Y Y Initiation initiatininitiatinininitiation ininininitiation initiatinininiti	5	woman, age undisclosed	March 2020	original wild type	Mar 2020	5 vaccines beginnin; in 2021		Х	Х	
7 30y.o. man May 2022, omicron, July 2022 July 2022 omicron July 2022 omicron July 2023 of 15 d) y -or individuals indicated as having short-term improvements bit not sustained improvements. this means that any improvements observed eventually dissipated. For case 4, he started valacyclovir one day before he started the long course of nimatreMir/itonavir, sufficuence of the benefits are due to immatreMir/itonavir, valacyclovir, or both. Table 2 Summary of cases who took an extend course of Paxlovid outside the context of an acute re-infection and did not experience any improvement	9	40 y.o. man	March 2020	original wild type	May 2020	Not specified	Sept 2022 (30 d)	×	×	
or individuals indicated as having short-term improvements but not sustained improvements, this means that any improvements observed eventually dissipated. For case 4, he started valecyclovir one day before he started the long course of nimatrelvir/ritonavir, stifficult to determine if the benefits are due to nimatrelvir/ritonavir, valecyclovir, or both.	7	30 y.o. man	May 2022, July 2022	omicron, omicron	Jul 2022	2021 (Mar, April, No		×		
Faxiovic			July 2020 v			2021 (Apr, May, Nov), 2022 (Jun. Oct)	Paxiovia Apr 2023 (10 d)			
35 y.o. July 2020 original Jul 2020 2021 (Apr, May, Nov), woman wild type 2022 (Jun, Oct)	0	34 y.o. woman	Nov 2021 0		2021	2021 (Oct, Nov)	May 2022 (15 d)			
July 2020 original Jul 2020 2021 (Apr, May, Nov), wild type Nov 2021 delta Nov 2021 2021 (Oct, Nov)	10	45 y.o. woman	Jan 2022 c			2021 (May, July), 2022 (Jan, Sept)	Nov 2022 (15 d)			
35 y.o. July 2020 original Jul 2020 2021 (Apr, May, Nov), 2022 (Jun, Oct) woman 34 y.o. Nov 2021 delta Nov 2021 2021 (Jun, Oct) 34 y.o. Nov 2021 delta Nov 2021 2021 (Jun, Oct) 000) 45 y.o. Jan 2022 omicron Jan 2022 2021 (May, July), 2022 woman woman Jan 2022 omicron Jan 2022 2021 (May, July), 2022 woman	11	37 y.o.	March 2020 c	original Mar 2020		2021 (Apr, Nov),	Feb 2023 (15 d attempted)			>

ŝ

Her symptoms of fatigue, orthostatic intolerance, and PEM worsened following a COVID-19 vaccine booster (Comirnaty) in October 2022.

In February 2023, she began a 15-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. Concomitant medications included low dose naltrexone, cetirizine, norethindrone acetate-ethinyl estradiol-iron, lumbrokinase, serrapeptase, and palmitoylethanolamide. During her nirmalterlvir/ritonavir course, she reported improvement in her energy and feeling stronger. Her blood pressure was consistently higher during this period (~145/80) and she did not experience orthostatic intolerance. However, she also experienced severe constipation and a severe flare of her MCAS symptoms (e.g., rashes, headaches, increased food intolerances). Two weeks after completing the nirmatrelvir/ritonavir course, she reported improved energy, physical strength, and mental clarity, and complete resolution of symptoms of orthostatic intolerance, which have not recurred as of July 2023. Although she had ongoing MCAS symptoms, these also improved with mast-cell stabilizers H1 and H2 blockers.

Case #3

In May 2022, a previously healthy 25-year-old woman with a medical history notable only for mild seasonal allergies developed COVID-19 symptoms. The infection was confirmed via rapid antigen test. She had previously received the primary two-shot series of the COVID-19 vaccine (Comirnaty) in April 2021, and a Comirnaty booster in November 2021. Initially, her Long COVID symptoms were primarily gastrointestinal, including lack of appetite, bloating, weight loss, and new food intolerances. Over the next few months, she developed additional symptoms, including fatigue, PEM, exercise intolerance, brain fog, headaches, noise and light sensitivity, and joint pain. In November and December 2022, she was treated for small intestinal bacterial overgrowth with two courses of rifaximin, which did not lead to sustained improvement; specifically, after the first course in November, her gastrointestinal symptoms improved for two days before returning to pre-rifaximin baseline, and after the second course in December, there was no improvement.

In January 2023, she began a 10-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. At the time, she was taking loratadine and a multivitamin. On day 4, she experienced a severe "crash," characterized by increased fatigue, headache, brain fog, sound and light sensitivity, and difficulty remaining upright, walking, and balancing. Approximately one month later, she had recovered from the initial crash and her fatigue and PEM had lessened meaningfully; she also experienced limited improvement in brain fog and headaches. In late February and March 2023, she switched from loratadine to fexofenadine and famotidine. In April and May 2023, the fatigue, PEM, brain fog, and headache benefits dissipated and her symptoms returned to her pre-nirmaltrevir/ritonavir baseline, and she began taking quercetin, low-dose naltrexone, and a probiotic. As of July 2023, she was experiencing some improvements in symptoms, was slowly reintroducing foods not tolerated earlier in her disease course, and added gentle swimming multiple times per week without PEM. She also began a 3-month course of valacyclovir with celecoxib, given their antiviral and anti-inflammatory properties, respectively, but observed no benefits.

Case #4

In August 2022, a 51-year-old man with a medical history notable only for tinnitus tested positive on PCR for SARS-CoV-2. He had previously received four doses of SARS-CoV-2 vaccines (Comirnaty in April and May 2021, Spikevax in November 2021 and August 2022), most proximally three weeks prior to COVID-19 symptom onset. His acute COVID symptoms included mild cough, fatigue, severe brain fog, heart rate variability, symptoms of paresthesia (skin crawling sensations and tingling in the back), visual snow and mild warping visual distortions, and loss of smell and taste. He began a 5-day course of nirmatrelvir/ritonavir the same day he tested positive (and within 1 day of experiencing COVID-19 symptoms), during which he experienced improvement in some symptoms. After completing the 5-day course, he tested negative on day five, then experienced rebound

Table (Table 3 Summary of cases who took an extended course	ases who took a	n extended cou	urse of Paxlovid v	of Paxlovid within the context of an acute re-infection	an acute re-infectio	Ę		
Case #	Case # Age, gender	Date of COVID infection(s)	Likely variant	Initiation of Long COVID symptoms	COVID vaccinations	Date and length of extended course of Paxlovid	Short-term improvement with Paxlovid	Sustained improvement with Paxlovid	Adverse effects of Paxlovid
12	41 y.o. woman	Dec 2020	alpha	Dec 2020	2021 (Mar, Apr), 2022 (Jan, Nov)	Apr 2023 (10 d during acute infection)	Apr 2023 (10 d during (returned to pre-reinfection acute infection) baseline within 2 months)		Y
13	55 y.o. transgender Mar 2020, Mar person 2021, Dec 2022	Mar 2020, Mar 2021, Dec 2022	original wild type, alpha, omicron	Mar 2020	2021 (Feb, July, Oct), 2022 (Oct), 2023 (Apr)	Dec 2022 (7.5 d during acute infection)	Dec 2022 (7.5 d during (returned to pre-reinfection acute infection) baseline within 3 weeks)		

symptoms, testing positive again 2-3 days after his last nirmatrelvir/ritonavir dose. This was associated with worsening cough, fatigue, headaches, brain fog, symptoms of dysautonomia, neuropathy, and inflammatory symptoms including joint pain and range of motion restriction. He tested every 2-3 days, continuing to test positive up until day 18. On day 19, he tested negative, although he continued to experience ongoing cough, brain fog, and exhaustion. He tested negative again on days 20, 22, and 24, then stopped testing regularly. His Long COVID symptoms included fatigue, daily headaches, brain fog, mental exhaustion, PEM, racing heart, palpitations, tinnitus, visual and sensory disturbances, sensitivity to noise, difficulty regulating his emotions, and paresthesia. He began taking fexofenadine, Alpha-Lipoic acid, Acetyl-L-carnitine, which improved energy and inflammatory symptoms some, and ibuprofen as needed for joint pain, which also seemed to occasionally help brain fog. These symptoms ebbed and flowed but got progressively better, until January 2023, when he experienced a severe crash after trying to exercise, from which he did not recover.

In March 2023, he started a 15-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. He concurrently began taking valacyclovir to address reactivation of Epstein-Barr virus (EBV Early Antigen D AB (IGG) levels were 143) and received a Glutathione/Vitamin-B complex shot two days before. He had never previously had EBV reactivation and was unaware of his original infection date, though antibody levels showed this was not his initial EBV infection. By the end of his first week taking valacyclovir and nirmatrelvir/ritonavir, his energy improved and he felt that he was back to his baseline health for several days. His PEM decreased in frequency, intensity, and duration, requiring him to rest for only 6-36 h to recover, instead of his usual 2 weeks. His symptoms of paresthesia resolved and he noted fewer headaches and improvement in his tinnitus, cognition, and affect. He continued to feel symptom reduction for four weeks after the completion of the nirmatrelvir/ritonavir and two weeks after the completion of the valacyclovir courses. After that point, he began to experience "crashes" again with more intensity. He then resumed valacyclovir, experiencing improvement within 24 h.

As of early June 2023, he reported feeling "radically better" and had returned to 90–95% of his previous health baseline five days per week and 75–85% the other two days. He had not yet returned to aerobic workouts, but was able to walk approximately five miles with a 30-pound backpack and do light weightlifting. He reported substantial improvement in his brain fog, mental exhaustion, affect, sensitivity to noise, symptoms of neuropathy, and ability to exert himself physically and complete resolution of his headaches, problems with regulating his emotions, visual disturbances, palpitations, and tachycardia. In mid-June 2023, he added low-dose naltrexone; since then, he has felt "much more stable" and has experienced no PEM. As of mid July 2023, he reported feeling back to pre-COVID baseline on all symptoms, other than symptoms he attributed to deconditioning. He is slowly reintroducing aerobic exercise.

Case #5

In March 2020, an adult woman (age not disclosed) with a medical history of recurrent sinus infections developed symptoms consistent with COVID-19. At the time, COVID tests were not available in her location, but she tested negative for flu and strep throat and a clinical diagnosis of COVID-19 infection was made. Her acute symptoms included a high fever, trouble breathing, and cognitive difficulties (e.g., not remembering how to use a phone to call for help). Her primary Long COVID symptoms included respiratory issues, fatigue, PEM, brain fog, and body aches. She had previously been able to exercise daily, but as a result of Long COVID was no longer able to. Laboratory testing was notable for a high anti-nuclear antibody titer at the beginning of her post-COVID symptoms (which later returned to normal), as well as the presence of EBV and Guillain-Barré Syndrome (GBS) antibodies, none of which had previously been documented in her medical records (she had previously tested normal for anti-nuclear antibodies and negative for EBV; she had not previously been tested

for GBS antibodies). She subsequently received five COVID-19 vaccines (Spikevax), beginning in 2021, at the recommended time intervals for primary series and boosters. Each of these vaccines led to some improvement in her symptoms. She also experienced improved respiratory symptoms after a pneumoccocal-23 vaccination. The only medication she takes is galcanezumab-gnlm for migraines.

In June 2023, she began a 10-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. Within a few days, her severe brain fog disappeared and her respiratory sensitivity and shortness of breath improved. She also saw meaningful improvement in multiple health metrics, including a lowering in respiratory rate, an improvement in oxygen saturation via pulse oximeter, and a lowering in heart rate (as measured with a wearable device), including during physical exertion activities. She described the cardiovascular improvement after nirmatrelvir/ ritonavir as "vast". For example, she compared her heart rate while doing cardiovascular exercise from three time points: prior to onset of her Long COVID symptoms, during Long COVID pre-nirmatrelvir/ritonavir, and during Long COVID post-nirmatrelvir/ritonavir, and her heart rate did not go up as much post-nirmatrelvir/ritonavir as pre-nirmatrelvir/ritonavir. She was also able to go on several work-related trips without crashing or exacerbating symptoms, which she had not been able to do since her Long COVID began.

Case #6

In March 2020, a 40-year-old man who was a prior competitive soccer player and mountain climber developed COVID-19 symptoms, including anosmia and shortness of breath. The infection was confirmed with an antibody test in April 2020. His respiratory and inflammatory symptoms persisted for ~10 days, then substantially resolved. Approximately eight weeks post-acute infection, in late May 2020, he began to experience plantar-palmar sensory neuropathies that gradually progressed in intensity and pain/paresthesias that were felt progressively more centrally, as well as a very high frequency of gastrointestinal symptoms, including frequent diarrhea and progressive food sensitivities. In September 2020, he began to develop blurry vision. He received the primary series of the COVID-19 vaccine in March and April 2021 (Comirnaty). After the first shot, he had a severe reaction (grapefruit-sized painful injection site reaction and flared symptoms) for five days, followed by 7 days of total symptom remission (clearer vision, neuropathic symptomology disappeared, return of energy), then a decline for three weeks. After the second shot, he experienced mild tinnitus for a day, but had no reaction at the injection site and no impact on symptoms.

He completed a 5-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection in the first quarter of 2022, leading to near-complete remission in symptoms by day 3, with a slow but progressive return of symptoms on day 6. He subsequently took a 30-day course of nirmatrelvir/ritonavir in September 2022, again outside of the context of an acute COVID-19 infection. During this course, he experienced slow remission of symptoms over the first 10 days, with progressive improvement in energy, vision clarity, and cognition over the subsequent weeks. Symptom improvement persisted for nearly two months, with a return of mild symptoms in February 2023. For example, he had had severe gastroesophageal reflux disease that developed as part of Long COVID, which disappeared upon taking nirmatrelvir/ritonavir, but has since returned. He was reinfected, as confirmed by rapid antigen test, during international travel in March 2023, and received a 10-day course of nirmatrelvir/ritonavir, which led to a rapid resolution of acute symptoms. Subsequent detailed serologic analysis of antibodies (Serimmune) in early April 2023 showed an absence of COVID-related antibodies, including those typically induced via vaccination. He was subsequently approved for intravenous immunoglobulin (IVIG), based, in part, on the presence of anti-trisulfated heparin disaccharide antibodies confirmed by multiple labs, and has had a progressive reduction in symptoms over the subsequent six months.

As of July 2023, his symptoms have improved somewhat, including less frequent waking from sleeping due to neuropathy, some brain fog, more

mild POTS symptoms, sporadic joint pain in hands, intermittent pain in one of his legs, milder headaches, and being able to add some foods back to his diet (which remains restricted by gastrointestinal symptoms, including diarrhea).

Case #7

A 29-year-old man had clinically confirmed SARS-CoV-2 infections in May 2022 and July 2022. His symptoms appeared to resolve fully after his first infection. After his second infection, his acute symptoms resolved after three weeks, at which time his Long COVID symptoms arose. His Long COVID symptoms include extreme fatigue, PEM, tachycardia, POTS, persistent headache, brain fog, chest pain and shortness of breath on exertion, and unrefreshing sleep; he met the myalgic encephalomyelitis/chronic fatigue syndrome (ME/CFS) diagnostic criteria²⁰. Since the onset of his Long COVID symptoms, he also developed constipation, chilblains, atopic dermatitis on his face, and paresthesia. His abnormal test results include high cholesterol and prominent Virchow spaces on a brain MRI. Before his initial infection, he had received three doses of the Comirnaty vaccine (March 2021, April 2021, and November 2021). He received an additional dose of the Spikevax vaccine in October 2022.

In May 2023, he began a 15-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. Concomitant medications included finasteride, amphetamine/dextroamphetamine salts, bupropion, atenolol, midodrine, Vitamin D3, and famotidine. Approximately six days into the course, he noticed slight ("~10%") improvement in his fatigue and brain fog. These improvements were sustained and increased to "~20%" by the end of the course. His 15 days on nirmatrelvir/ritonavir also coincided with the longest number of consecutive days (15) without any worsening of symptoms (e.g., a crash). However, two days after the final nirmatrelvir/ritonavir dose, he experienced a PEM crash, which included a sore burning sensation in his arms, poor sleep, headache, fatigue, and brain fog so intense that sitting, moving, and talking was difficult. He returned to his pre-nirmatrelvir/ritonavir Long COVID baseline within five days. He has remained at his pre-nirmatrelvir/ritonavir Long COVID baseline in the subsequent months.

Case #8

In July 2020, a 35-year-old woman who was a prior long-distance runner developed a clinically confirmed SARS-CoV-2 infection. While she was not hospitalized during her acute COVID infection, she experienced loss of taste and smell, fever, sore throat, fatigue, and substantial difficulty breathing. After the acute phase, she continued to experience symptoms consistent with Long COVID, including fatigue, PEM, and migraines, with her migraines manifesting as the most debilitating aspect of her PEM. Her difficulty breathing slowly resolved over six months; she experienced chest pain starting two months after the acute infection and largely resolving over about two years. As part of her Long COVID symptoms, she was diagnosed with mild ME/CFS in January 2021, which has since progressed to moderate ME/CFS (as defined by ICC criteria²¹). Since her infection, she has received five COVID vaccinations (Comirnaty primary series in April and May 2021, Spikevax booster in November 2021, Comirnaty boosters in June 2022 and October 2022).

In April 2023, she initiated an extended course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. Concomitant medications included acetaminophen, sumatriptan, Excedrin, ubuiquinone, taxifolin, a multivitamin, Vitamins C, D, and E, and zinc. For the first 10 days, she took half doses of nirmatrelvir/ritonavir (spreading a 5-day course over 10 days). For the subsequent five days, she took full doses of nirmatrelvir/ritonavir, completing an additional 5-day course. During this time, she experienced some insomnia and taste-related side effects, but did not observe any change in her Long COVID symptoms.

Case #9

In November 2021, a 34-year-old-woman developed symptoms consistent with a COVID-19 infection, confirmed by rapid antigen test. She had previously received two doses of the Comirnaty SARS-CoV-2 vaccine in October and November 2021. Following the acute illness, she felt herself to have recovered, with no lingering symptoms other than a feeling of being "wired." Approximately one week after the acute infection, she went for a run and experienced a first episode of PEM 24 h later that caused her to collapse, and she was hospitalized as a result. Extensive laboratory testing was notable only for mild anemia. This episode of PEM precipitated three months of symptoms consistent with severe ME/CFS in which she was largely bedbound and dependent on others for care. During this time, she experienced fatigue, brain fog, short-term memory loss, and difficulty finding and remembering words. Her symptoms also met the diagnostic criteria for dysautonomia, POTS, nerve damage, and small fiber neuropathy. Self-treatment with cannabinoid resulted in some improvement, which shifted her from severe to moderate ME/CFS (mostly housebound, by ICC criteria)²¹.

In May 2022, she took a 15-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection. Her only concomitant medication was aspirin. She did not experience any change in her symptoms during or after the nirmatrelvir/ritonavir course. Since that time, her doctor has treated her with rapamycin, which has led to substantial relief in her ME/ CFS-related symptoms.

Case #10

In January 2022, a 45-year-old woman developed symptoms of a COVID-19 infection. Her Long COVID symptoms included fatigue, breathing difficulties, and severe chest pain diagnosed as costochondritis. Over the first eight weeks of her illness, her Long COVID symptoms worsened and she experienced severe weight loss and migraines. Since that time, clinicians diagnosed her with chronic pericarditis, POTS, cardiac fibrosis, ME/CFS, and possible MCAS. Before her illness, she had received two doses of the Comirnaty vaccine (May and July 2021) and a dose of the Spikevax vaccine in January 2022 (immediately prior to her diagnosis). She received an additional dose of the Spikevax vaccine in September 2022.

In October 2022, she completed a 5-day course of nirmatrelvir/ritonavir in the setting of a SARS-CoV-2 reinfection. She experienced three days with no fatigue and in which she described "feeling normal." However, these improvements did not persist, and she returned to her pre-reinfection baseline. Because of this temporary improvement, she completed a second, 15-day course of nirmatrelvir/ritonavir in late November 2022 outside of the context of an acute COVID-19 infection. This additional course of nirmatrelvir/ritonavir did not improve her fatigue or her other symptoms in any way.

Case #11

A 33-year-old woman developed symptoms of a SARS-CoV-2 infection in March 2020 (clinically confirmed via emergency room telehealth). For the first two weeks of her illness, her symptoms included sore throat and excessive daytime fatigue and sleepiness. Shortly thereafter, her condition worsened considerably and she felt severely ill for another 3-4 weeks. This left her bedridden, only able to go to the bathroom on her own (which required resting before crawling back to bed), and a similar crawling and resting routine to get any food 1-2 times daily. Her symptoms included extreme fatigue and PEM with myalgia, sensory sensitivity and flu-like symptoms lasting days or weeks; POTS with dizziness, temperature irregularity, and low blood pressure; insomnia and unrefreshing sleep; joint and muscle pain exacerbated by PEM; gastrointestinal issues with alternating constipation and diarrhea; no libido; and cognitive issues affecting concentration, word finding, memory, and information processing. While she experienced some symptom improvement over time, other symptoms (e.g., fatigue, confusion, headache, myalgia) increased in severity through spring 2021, and she became bedridden again in spring 2022. Since her initial infection, she received the Jcovden vaccine in April 2021, a dose of the Comirnaty vaccine in November 2021, and a dose of the Nuvaxovid vaccine in November 2022.

In May 2022, she took a 5-day course of nirmatrelvir/ritonavir outside of the context of an acute COVID-19 infection, experiencing no side effects other than dysgeusia. This course led to improvements in overall functioning, such that she was no longer bedridden and other symptoms were less severe (e.g., fatigue improved, migraine improved, light and sound sensitivity reduced enough to be able to leave dark bedroom for short periods of time, shifted from needing to be fed while lying down to being able to sit up in bed to eat and drink). Concomitant medications included escitalopram, propranolol, low-dose naltrexone, pravastatin, fenofibrate, and maraviroc. In June 2022, she attempted a second 5-day course of nirmatrelvir/ritonavir, also outside of the context of an acute COVID-19 infection. She experienced severe stomach pain within 5 h of taking the first dose and discontinued the course. This stomach pain lasted 7-9 h and she had gastrointestinal issues for the next week. At the time of the second course, concomitant medications included those listed above, as well as clopidogrel and meloxicam. Suspecting that the stomach pain was due to contraindicated medications, she attempted a third, 15-day course of nirmatrelvir/ritonavir in February 2023, also outside of the context of an acute COVID-19 infection. At this time, she stopped all of her other medications except escitalopram, propranolol, and low-dose naltrexone. She again experienced severe stomach pain within 5 h of the first dose of nirmatrelvir/ ritonavir and stopped the course. As before, her stomach pain lasted 7-9 h, with gastrointestinal issues lingering for the next week.

The improvements from the initial 5-day course of nirmatrelvir/ritonavir have persisted, but she continues to experience her same Long COVID symptoms and substantial functional limitations.

Case #12

In December 2020, a 41-year-old physically fit woman who ran marathons developed symptoms of a COVID-19 infection, confirmed by a positive PCR test. During the acute phase of her COVID infection, she experienced body aches, dizziness, vertigo, gastrointestinal issues, difficulty breathing, and chest pain. After the acute phase, her dizziness, vertigo, chest pain, and difficulty breathing continued and she developed widespread tingling and numbness, sensitivity to bright lights and loud noises, and PEM. She was unable to return to work for nine months. Since her initial infection, she has had four shots of the Spikevax COVID-19 vaccine: her primary series in March/April 2021, and boosters in January 2022 and November 2022.

She experienced slow improvements in her symptoms starting six months after initial infection. Her vertigo, difficulty breathing, gastrointestinal issues, and nose bleeds improved in the summer of 2021. She used pacing as a strategy to avoid PEM, and returned to work in September 2021. Throughout 2022, her symptoms slowly improved and she found daily stretches helped with the chest pain. By February 2023, her only remaining Long COVID symptom was chest pain, managed by the daily stretching. She had returned to running and lifting weights again.

In April 2023, she was reinfected with SARS-CoV-2 and was prescribed a 10-day course of nirmatrelvir/ritonavir. Concomitant medications included zinc, vitamin C, vitamin D, astragalus, echinacea, and melatonin. She had also taken one dose of ivermectin three days prior to starting nirmatrelvir/ritonavir. On the seventh day (after 13 doses) of nirmatrelvir/ ritonavir, she experienced itchiness and tingling throughout her body (legs, abdomen, neck) and in her mouth and throat. She discontinued the course and her itchiness resolved by the next morning. For approximately the next two weeks, she experienced dizziness, and shakiness and tingling in her arms and hands. She took Tailwind Endurance Fuel to ease these symptoms. Two weeks after stopping the nirmatrelvir/ritonavir, she had returned to her previous baseline. However, within another two days, she developed a new sore throat and experienced tingling and dizziness. She then tested positive again for SARS-CoV-2 on a rapid antigen test. She completed the remaining three and a half days (seven doses) of nirmatrelvir/ritonavir, noting improvement in her symptoms. One week after finishing nirmatrelvir/ ritonavir, she continued to experience tingling and dizziness and brain fog and had not returned to her previous baseline. Two months later, she had returned to her pre-April 2023 reinfection baseline.

Case #13

A 55-year-old transgender person assigned female at birth developed symptoms of a clinically confirmed COVID-19 infection in March 2020 and a reinfection in March 2021 (confirmed by PCR test). Before his illness, he had been diagnosed with Neuromyelitis Optica Spectrum Disorder (2011), ME/CFS (diagnosed in 2017, but began earlier), fibromyalgia, and psoriasis. Following his first COVID-19 infection, he developed tachycardia, brain fog, and mild cognitive symptoms. After his second COVID-19 infection, he developed MCAS symptoms, including frequent, mild-to-moderate sinus congestion, flushing, and throat swelling in relation to many foods. Currently, his Long COVID symptoms include brain fog, difficulty with word finding, MCAS symptoms, sensitivity to histamines, and intermittent PEM. He received one dose of the Spikevax vaccine in February 2021, two doses of the Jcovden vaccine in July and October 2021, a Comirnaty bivalent booster in October 2022, and one primary series dose of the Nuvaxovid vaccine in April 2023.

In December 2022, he took a 7.5 day course of nirmatrelvir/ritonavir during a third COVID-19 infection (confirmed by at-home molecular Lucira test and multiple rapid antigen tests). At the time, he was taking cromolyn sodium, ketitofen, low-dose naltrexone, metoprolol succinate XL, rosuvastatin, nortriptyline, modafinil, finasteride, loratadine, valacyclovir, famotidine, and nitrofurantoin daily; testosterone cypionate weekly; leuprolide quarterly; and Zofran and clonazepam as needed. His daily supplements included Deplin (algal oil), cranberry extract, laxin forte, vision optimizer blend, biotin, ubiquinol, alpha-lipoic acid, probiotics, fish oil, flaxseed oil, quercetin, vitamin C, acetyl-l-carnitine, milk thistle, P5P, calcium, skullcap and passionflower tinctures, and DAO and digestive enzymes. He also received twice-weekly injections of glutathione and methylcobalamin. During his acute COVID infection, he also took melatonin and high dose Vitamin C powder.

After this course of nirmatrelvir/ritonavir, he experienced several days with no fatigue or notable pain, although his fatigue and pain then returned to his pre-infection baseline within 2–3 weeks. This pattern resembled his experience with his first COVID infection, in which he experienced improvements in his fatigue and chronic pain during the ten days of initial infection, while also experiencing intermittent low-grade fevers. At that time, as his fevers subsided, his fatigue and pain returned.

Discussion

This case series describes 13 individuals with established Long COVID who reported variable outcomes following extended antiviral treatment with nirmatrelvir/ritonavir (summarized in Tables 1, 2, and 3). These individuals contracted Long COVID from different strains of SARS-CoV-2, and took nirmatrelvir/ritonavir at different lengths of time since their initial infection; we posit that it is a plus that our case series population reflects the diversity of people with Long COVID. Although the treatment was generally welltolerated, these individuals exhibited heterogeneity in the clinical manifestations of Long COVID and the impact of antiviral therapy on their symptoms; only 5 of the 13 cases experienced sustained improvement. Given the lack of accepted treatments for Long COVID, these cases contribute to burgeoning research suggesting that at least some individuals may benefit from antiviral therapy beyond the acute phase of infection. Taken together, this case series provides a strong rationale for the study of extended courses of nirmatrelvir/ritonavir and/or other antiviral medications in randomized clinical trials, some of which are underway, with the goal of informing clinical recommendations for using antivirals as a potential treatment option for Long COVID.

We acknowledge several limitations of this case series. First, while this study is large for a case series, it is still only reporting on the experiences of 13 individuals, who may or may not be representative of the experiences of all patients with Long COVID. For example, they were all at least somewhat networked because they participated in public fora for people with Long COVID and/or knew others with Long COVID. Additionally, all participants had technological savvy to communicate over the internet or phone, and were all proficient in the English language. Second, while the patient-led

nature of the study was a strength, we were also limited to patient-reported experiences, which could have been subject to recall bias or fatigue bias. However, given that participants often reported their experiences contemporaneously or near contemporaneously, we were hopefully able to minimize recall bias. Additionally, participants may have chosen to disclose more or fewer details based on their levels of fatigue; by being flexible about how they shared their experiences, we hopefully reduced any potential fatigue bias as well. Third, there was substantial heterogeneity in the data, including differing ages and symptoms of participants and varying lengths of time taking nirmatrelvir/ritonavir; this allowed us to document an illustrative breadth of experiences but did not provide the depth to understand the particular experiences of any given subgroup.

Nevertheless, our study had substantial strengths. To the best of our knowledge, this is the first case series of people with Long COVID who have attempted extended courses of nirmatrelvir/ritonavir, with most reporting their experiences outside of the context of an acute COVID-19 infection. The heterogeneous composition of cases is illustrative of the heterogeneity of experiences of patients with Long COVID. A major strength of this study is its grounding in patient-led and participatory approaches to draw from knowledge outside of traditional medical institutions. While we did not conduct independent reviews of medical records, the patients' own descriptions of their experiences provide insights that may otherwise not be captured due to limitations in medical record systems, including inaccuracies in recording symptoms and variability in healthcare system contact²². This is particularly the case with emerging illnesses like Long COVID that did not have its own ICD-10 diagnostic code until October 2021²¹, 1.5 years after some of the participants in this case series began experiencing Long COVID. What we report here- the lived experience of people with Long COVID who were able to access this treatment and report on its perceived impact- continues to build upon efforts by people experiencing Long COVID to describe and study the condition, which has been critical to driving forward this field of research. This approach may be useful for studying potential treatments for other infection-associated chronic conditions as well.

There is currently debate as to whether a 5-day duration of nirvmatrelvir/ritonavir therapy is sufficient in all cases during acute SARS-CoV-2 infection. In addition, some individuals initially reporting improvement in Long COVID symptoms have described a recurrence of symptoms following the 5-day course. For this reason, there is growing interest in the safety, tolerability, and efficacy of longer courses of antivirals like nirmatrelvir/ritonavir. Although the mechanism of persistence is different, there is precedent for longer courses of antivirals in the protease inhibitor class in both HIV²³ and hepatitis C²⁴.

There are several considerations for future research assessing antivirals in established Long COVID. First, there is growing consensus that there are likely to be different endotypes of Long COVID, which may be driven by different underlying biological processes. Ignoring this heterogeneity may make it difficult to identify a signal, if persistent viral replication drives only some cases of Long COVID. For this reason, clinical trials should further explore and attempt to phenotype individuals who exhibit a benefit, even if this signal is lost in the larger group of participants. In larger studies, we recommend enrolling adequate numbers of different phenotypes (by symptom clusters for now, until more specific phenotypes are identified), as is being attempted in at least one study of nirmatrelvir/ritonavir (NCT05595369), and assessing relevant outcomes accordingly. Second, prolonged courses of these agents may be worth further study. In this case series, some individuals experienced improvement only after ≥10 days of treatment. It will be important to trial a variety of lengths of time in order to determine if sustained resolution of symptoms is possible with longer courses. Third, some individuals experienced recurrence of previously improved symptoms following discontinuation of antiviral treatment. This suggests that researchers and regulators should consider the potential impact of treatment completion and determine whether ongoing treatment could be implemented to limit this symptomatic rebound. Fourth, although

treatment appeared to overall be safe in these individuals, some participants experienced flares of MCAS-related symptoms and/or post-exertional malaise. Such symptoms must be monitored closely to determine whether they are specifically related to antiviral therapy, and if so, this risk must be considered in addition to the known potential risks related to drug-drug interactions and other adverse effects.

If clinical trials do indeed detect a signal, there should be a clear regulatory pathway to ensure that these drugs can be made accessible to patients, especially since there are currently no FDA-approved treatments specifically for Long COVID. This might entail consideration of options like expanded access, fast-track designation, and breakthrough therapy designation to both encourage industry participation and increase access of pharmaceutical treatments to Long COVID patients.

In conclusion, while the case series findings are heterogeneous, we describe the use of long courses of Paxlovid as a potentially promising pharmaceutical treatment to support at least some people with Long COVID. These cases provide strong rationale for the ongoing study of antivirals for Long COVID to determine if, when, and how they should be used in this patient population.

Data availability

To protect the anonymity of participants, data are not publicly available.

Received: 7 December 2023; Accepted: 6 November 2024; Published online: 06 January 2025

References

- Altmann, D. M., Whettlock, E. M., Liu, S., Arachchillage, D. J., Boyton, R. J. The immunology of long COVID. *Nat. Rev. Immunol.* Published online July 11, 2023. https://doi.org/10.1038/s41577-023-00904-7 (2023).
- Davis, H. E., McCorkell, L., Vogel, J. M. & Topol, E. J. Long COVID: major findings, mechanisms and recommendations. *Nat. Rev. Microbiol.* **21**, 133–146 (2023).
- WHO Coronavirus (COVID-19) Dashboard. Accessed August 23, 2023. https://covid19.who.int (2023).
- Ballering, A. V., van Zon, S. K. R., olde Hartman, T. C. & Rosmalen, J. G. M. Persistence of somatic symptoms after COVID-19 in the Netherlands: an observational cohort study. *Lancet* 400, 452–461 (2022).
- Ford, N. D. Long COVID and Significant Activity Limitation Among Adults, by Age — United States, June 1–13, 2022, to June 7–19, 2023. MMWR Morb Mortal Wkly Rep. 72 https://doi.org/10.15585/mmwr. mm7232a3 (2023).
- Thaweethai, T. et al. Development of a definition of postacute sequelae of SARS-CoV-2 infection. JAMA 329, 1934–1946 (2023).
- Davis, H. E. et al. Characterizing long COVID in an international cohort: 7 months of symptoms and their impact. *eClinicalMedicine*. 38. https://doi.org/10.1016/j.eclinm.2021.101019 (2021).
- Swank, Z. et al. Persistent circulating severe acute respiratory syndrome coronavirus 2 spike is associated with post-acute coronavirus disease 2019 sequelae. *Clin. Infect. Dis.* **76**, e487–e490 (2023).
- Liotti, F. M. et al. Assessment of SARS-CoV-2 RNA test results among patients who recovered from COVID-19 with prior negative results. *JAMA Intern Med.* 181, 702–704 (2021).
- Vibholm, L. K. et al. SARS-CoV-2 persistence is associated with antigen-specific CD8 T-cell responses. *EBioMedicine* 64, 103230 (2021).
- Sun, J. et al. Prolonged persistence of SARS-CoV-2 RNA in body fluids - Volume 26, Number 8—August 2020 - *Emerging Infectious Diseases journal* - CDC. https://doi.org/10.3201/eid2608.201097 (2020).

- Proal, A. D. et al. SARS-CoV-2 reservoir in post-acute sequelae of COVID-19 (PASC). *Nat Immunol.* Published online September 4, 2023. https://doi.org/10.1038/s41590-023-01601-2 (2023).
- Proal, A. D., VanElzakker, M. B. Long COVID or post-acute sequelae of COVID-19 (PASC): an overview of biological factors that may contribute to persistent symptoms. *Front. Microbiol.* 12. https://www. frontiersin.org/articles/10.3389/fmicb.2021.698169 (2021).
- Wang, Y. et al. Early administration of Paxlovid reduces the viral elimination time in patients infected with SARS-CoV-2 Omicron variants. *J. Med Virol.* 95, e28443 (2023).
- Niraj, N., Mahajan, S. S., Prakash, A., Sarma, P. & Medhi, B. Paxlovid: a promising drug for the challenging treatment of SARS-COV-2 in the pandemic era. *Indian J. Pharmacol.* 54, 452 (2022).
- Xie, Y., Choi, T. & Al-Aly, Z. Association of treatment with nirmatrelvir and the risk of post–COVID-19 condition. *JAMA Intern Med.* 183, 554–564 (2023).
- Durstenfeld, M. S. et al. Association of Nirmatrelvir/Ritonavir treatment with long COVID symptoms in an online cohort of nonhospitalized individuals experiencing breakthrough SARS-CoV-2 infection in the omicron era. Published online March 5, 2023:2023.03.02.23286730. https://doi.org/10.1101/2023.03.02. 23286730 (2023).
- Peluso, M. J. et al. Effect of oral nirmatrelvir on long COVID symptoms: 4 cases and rationale for systematic studies. *Pathog. Immun.* 7, 95 (2022).
- Geng, L. N., Bonilla, H. F., Shafer, R. W., Miglis, M. G., Yang, P. C. Case report of breakthrough long COVID and the use of nirmatrelvirritonavir. Published online 2022.
- IOM 2015 Diagnostic Criteria | Diagnosis | Healthcare Providers | Myalgic Encephalomyelitis/Chronic Fatigue Syndrome (ME/ CFS) | CDC. Published April 27, 2021. Accessed August 24, 2023. https://www.cdc.gov/me-cfs/healthcare-providers/diagnosis/iom-2015-diagnostic-criteria.html.
- 21. Carruthers, B. M. et al. Myalgic encephalomyelitis: international consensus criteria. *J. Intern Med.* **270**, 327–338 (2011).
- 22. Zhang, H. G. et al. Potential pitfalls in the use of real-world data for studying long COVID. *Nat Med*. Published online 2023:1-2.
- Gandhi, R. T. et al. Antiretroviral drugs for treatment and prevention of HIV infection in adults: 2022 recommendations of the international antiviral society–USA panel. *JAMA* 329, 63–84 (2023).
- 24. Bhattacharya, D., Aronsohn, A., Price, J., Lo Re V. III, the American Association for the Study of Liver Diseases–Infectious Diseases Society of America HCV Guidance Panel. Hepatitis C Guidance 2023 Update: American Association for the Study of Liver Diseases– Infectious Diseases Society of America Recommendations for Testing, Managing, and Treating Hepatitis C Virus Infection. *Clin. Infect Dis.* Published online May 25, ciad319. https://doi.org/10.1093/ cid/ciad319 (2023).

Acknowledgements

We appreciate the support of all individuals who shared their experiences for inclusion in our study. Dr. Peluso is supported on K23AI157875.

Author contributions

A.K.C. originated the research, in consultation with M.P. and L.M.; A.K.C. led the writing, in collaboration with T.W.J., E.M.S., L.K., and S.S.; A.K.C. led the data collection, in collaboration with J.M.V., J.H-G., D.L., S.C., K.L., S.C.B., G.L., J.D.D., and L.M. All authors had the opportunity to review and revise the manuscript and approve the submitted version. As corresponding author, A.K.C. certifies that the data were collected transparently and preserved, that the data presented accurately reflect the original data, and that all authors have certified the author list and author contributions.

Competing interests

Dr. Peluso has served as a consultant for Gilead Sciences and AstraZeneca, outside the submitted work. The remaining authors have no conflicts of interest to declare.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-024-00668-8.

Correspondence and requests for materials should be addressed to Alison K. Cohen.

Peer review information *Communications Medicine* thanks Seth Congdon and the other, anonymous, reviewer(s) for their contribution to the peer review of this work.

Reprints and permissions information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License, which permits any non-commercial use, sharing, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if you modified the licensed material. You do not have permission under this licence to share adapted material derived from this article or parts of it. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/bync-nd/4.0/.

© The Author(s) 2025